Author's response to reviews

Title: Eustachian valve endocarditis: a rare localization of right side endocarditis. A case report and review of the literature

Authors:

Adriano M Pellicelli (apellicelli@scamilloforlanini.rm.it)
Paolo Pino (ppino@scamilloforlanini.rm.it)
Antonio Terranova (aterranova@scamilloforlanini.rm.it)
Cecilia D'Ambrosio (cdambrosio@scamilloforlanini.rm.it)
Fabrizio Soccorsi (fsoccorsi@scamilloforlanini.rm.it)

Date: 13 July 2005

Author's response to reviews:

1. we have enclosed in the text the cases of San Roman et al.
2. We have discussed the features of San Roman patients and we have enclosed all the cases in the Table.
3. we don't have any images after antibiotic therapy because the patient, a drug abuser, was lost to the follow up.
4. We believe that pathogenetic mechanism of infection could be of interest for the readers and could be enclosed in the present case report in accord to the reviewer.
5. we cannot include any clip of the case reported
6. We have done all the requested formatting changes as reported in the technical review 111.